Ther Adv Hematol

2021, Vol. 12: 1-9 DOI: 10.1177/ 20406207211048361

© The Author(s), 2021. Article reuse guidelines: sagepub.com/journalspermissions

#### Correspondence to: Monica Carpenedo

Hematology and Transplantation Unit, ASST San Gerardo Hospital, Via G. B. Pergolesi, 33, 20900 Monza MB, Italy. mnc.carpenedo@gmail.

# com

Erminia Baldacci Department of Hematology, Policlinico Umberto I University Hospital, Rome, Italy

### Claudia Baratè

Hematology and Bone Marrow Transplant Unit, Department of Clinical and Experimental Medicine, University of Pisa, Pisa, Italy

Alessandra Borchiellini Regional Reference Center of Haemostasis and Thrombosis Haematology Unit, Città della Salute e della Scienza, Turin, Italy

### Francesco Buccisano

Hematology, Department of BioMedicine and Prevention, University of Rome "Tor Vergata", Rome, Italy

### Giuseppina Calvaruso

Campus of Hematology Franco and Piera Cutino, AOOR Villa Sofia-V. Cervello, Palermo, Italy

### Federico Chiurazzi

Department of Clinical Medicine, University "Federico II" of Naples, Naples, Italy

## Bruno Fattizzo

UOC Ematologia, Fondazione IRCCS Ca' Granda Ospedale Maggiore Policlinico, Milan, Italy

Department of Oncology and Oncohematology, University of Milan, Milan, Italy

### Gaetano Giuffrida

Division of Haematology, A.O.U. Policlinico Vittorio Emanuele, Catania, Italy

Second-line administration of thrombopoietin receptor agonists in immune thrombocytopenia: Italian Delphi-based consensus recommendations

Monica Carpenedo<sup>(D)</sup>, Erminia Baldacci, Claudia Baratè, Alessandra Borchiellini, Francesco Buccisano, Giuseppina Calvaruso, Federico Chiurazzi, Bruno Fattizzo, Gaetano Giuffrida, Elena Rossi, Francesca Palandri, Potito Rosario Scalzulli, Sergio Mario Siragusa, Angelantonio Vitucci and Francesco Zaja

# Abstract

**Introduction:** In patients with primary immune thrombocytopenia (ITP), a short course of steroids is routinely given as first-line therapy. However, the response is often transient and additional therapy is usually needed. Thrombopoietin receptor agonists (TPO-RAs) are frequently used as second-line therapy, although there is little clinical guidance on the timing of their administration and on tapering/discontinuation of the drug. To provide clinical recommendations, we used the Delphi technique to obtain consensus for statements regarding administration and on tapering/discontinuation of second-line TPO-RAs among a group of Italian clinicians with expertise in management of ITP.

**Methods:** The Delphi process was used to obtain agreement on five statements regarding initiation and on tapering/discontinuation of second-line TPO-RAs. Agreement was considered when 75% of participants approved the statement. Eleven experts participated in the voting. **Results:** Full consensus was reached for three of the five statements. The experts held that an early switch from corticosteroids to a TPO-RA has the dual advantage of sparing patients from corticosteroid abuse and improve long-term clinical outcomes. All felt that dose reduction of TPO-RAs can be considered in patients with a stable response and platelet count >100 × 10<sup>9</sup>/L that is maintained for at least 6 months in the absence of concomitant treatments, although there was less agreement in patients with a platelet count >50 × 10<sup>9</sup>/L. Near consensus was reached regarding the statement that early treatment with a TPO-RA is associated with an increase in clinically significant partial or complete response. The experts also agreed that optimization of tapering and discontinuation of TPO-RA therapy in selected patients can improve the quality of life.

**Conclusion:** The present consensus can help to provide guidance on use of TPO-RAs in daily practice in patients with ITP.

# Plain language summary

# Second-line administration of thrombopoietin receptor agonists in immune thrombocytopenia

• There is little guidance on the timing of administration and tapering/discontinuation of thrombopoietin receptor agonists (TPO-RAs) in patients with primary immune thrombocytopenia (ITP).

journals.sagepub.com/home/tah



Creative Commons Non Commercial CC BY-NC: This article is distributed under the terms of the Creative Commons Attribution-NonCommercial 4.0 License (https://creativecommons.org/licenses/by-nc/4.0/) which permits non-commercial use, reproduction and distribution of the work without further permission provided the original work is attributed as specified on the SAGE and Open Access pages (https://us.sagepub.com/en-us/nam/open-access-at-sage).

Elena Rossi

Department of Radiological and Hematological Sciences, Universita Cattolica del Sacro Cuore and Fondazione Policlinico Universitario A. Gemelli IRCCS, Rome, Italy

Francesca Palandri IRCCS Azienda Ospedaliero-Universitaria di Bologna, Istituto di Ematologia "Seràgnoli", Bologna, Italy

Potito Rosario Scalzulli Department of Hematology and Stem Cell Transplant Unit, Fondazione IRCCS "Casa Sollievo della Sofferenza", San Giovanni Rotondo, Italy

Sergio Mario Siragusa Department PROMISE, University of Palermo, Palermo, Italy

Angelantonio Vitucci Haematology with Transplant Unit, Az. Universitario-Ospedaliera Consorziale Policilinico, Bari, Italy

Francesco Zaja Department of Medical, Surgical and Health

Sciences, University of Trieste, Trieste, Italy Hematology, Azienda

Sanitaria Universitaria Giuliano Isontina, Trieste, Italy

- The Delphi technique was used to obtain consensus for five statements.
- The present consensus among Italian clinicians aims to provide guidance on secondline use of TPO-RAs for patients with ITP in daily practice.

*Keywords:* consensus, Delphi, immune thrombocytopenia, management, second line, therapy, thrombopoietin receptor agonists

Received: 16 April 2021; revised manuscript accepted: 6 September 2021.

## Introduction

Primary immune thrombocytopenia (ITP) is an acquired autoimmune disorder characterized by reduced platelet counts related to increased peripheral destruction of platelets and diminished production.<sup>1</sup> ITP may have strong negative impact on the patient's quality of life, which is mainly related to the significant burden of therapy, bleeding events, and fatigue.<sup>2</sup> Diagnosis of primary ITP is usually made following exclusion of underlying and/or precipitating causes of the thrombocytopenia and is classified based on duration as newly diagnosed, persistent (3–12 months), and chronic ( $\geq$ 12 months).<sup>3</sup> In adults, ITP often has a chronic course.<sup>2,4</sup>

The main goals of therapy are to terminate any active bleeding events and to limit future hemorrhage. In a newly diagnosed patient with ITP, the guidelines of the American Society of Hematology (ASH) suggest that patients with a platelet count of  $<30 \times 10^{9}$ /L and asymptomatic should be treated with first-line corticosteroids [prednisone (0.5–2.0 mg/kg per day) or dexamethasone (40 mg per day for 4 days)].<sup>5</sup>

Even if the majority of patients will initially respond to corticosteroids, unfortunately the response is often transient and additional therapy is needed.6 Repeated or sustained administration of corticosteroids is not recommended given the potential for adverse effects and negative impact on quality of life such as weight gain, hypertension, osteoporosis, hyperglycemia, mood alterations, and infections.<sup>5,7</sup> Moreover, prolonged treatment will not improve the patient's response, but will likely worsen adverse events. The high rates of recurrence further demonstrate that corticosteroids reduce symptoms, but do not change the course of the disease.8 Importantly, all guidelines and consensus recommendations suggest a short course of corticosteroid therapy ( $\leq 6$  weeks) and limited to 1-2 cycles, while carefully monitoring for adverse effects.<sup>3,5,7–9</sup> However, the administration of corticosteroids is not always in line with these recommendations. For example, in the 472 physicians from 13 countries participating in I-WISh (ITP World Impact Survey), 75% referred that they commonly prescribe corticosteroids or steroids at first relapse.<sup>10</sup> In addition, in a German survey of 1023 patients, 62% of patients receiving steroids as first-line treatment therapy received them for up to 6 months, and 29% received them for more than a year during second- and even third-line treatment.<sup>11</sup>

In many patients, ITP becomes persistent or chronic, and second-line treatment may thus be warranted.<sup>5,12,13</sup> In this regard, it should be considered that limiting corticosteroids to the first line of therapy allows earlier transition to secondline therapy. Second-line therapy should have the same purpose as first-line treatment and aim to achieve long-term remission. The standard options for second-line therapy comprise splenectomy, thrombopoietin receptor agonists (TPO-RAs), and rituximab.<sup>5</sup> Splenectomy has been historically considered as the main option for second-line management of ITP given its potential to provide long-term remission. Splenectomy is associated with high initial response rate, although about one-third of patients will experience relapse over long-term follow-up and there are no reliable predictors of long-term response.14 Moreover, splenectomy is associated with potentially serious long- and short-term risks due to loss of hematological and immunological functions.<sup>14,15</sup> These include reduced Fcy R-mediated opsonization of microorganisms, reduced interaction of memory B-cells and helper T-cells, impairment of contact between abnormal particulates, and a densely populated reticulum of macrophage.<sup>16</sup> Splenectomy is also associated with an increased risk of thrombosis, with the highest incidence of venous thromboembolism in the perioperative period and during the first year after the procedure with a cumulative prevalence ranging from 1.4% to 16% depending on the length of follow-up. $^{17}$ 

Rituximab has been associated with long-term remission in a proportion of patients, although there are no clear indications on which patients and when it should be administered.<sup>18</sup> Moreover, given the current CoOVID-19 pandemic, there are also concerns about the use of rituximab since patients receiving it will be unlikely to make antibodies to SARS-CoV-2.<sup>19</sup>

TPO-RAs are often the preferred agent for second-line medical therapy, and some authors have suggested postponing splenectomy as long as possible, or at least up to 1 year.<sup>14</sup> Indeed, a recent consensus approach from a Spanish group reported that 97.5% of these experts consider TPO-RAs as the class of choice for second-line therapy of patients with ITP.9 TPO-RAs also have the most evidence demonstrating their longterm effectiveness.<sup>5,7–9</sup> While there is convincing consensus on the use of TPO-RAs as second-line therapy, the optimal timing for their administration has not been universally defined. The 2019 ASH guidelines, for example, state that TPO-RAs can only be considered for ITP lasting  $\geq$ 3 months in adults.<sup>5</sup>

Another clinically relevant aspect regards tapering of TPO-RAs. Tapering represents a key issue since up to 30% of patients with ITP maintain a sustained response after dose reduction or discontinuation of the TPO-RA.<sup>3</sup> Indeed, a scheme has recently been proposed for tapering and discontinuation of TPO-RA.20 However, this is limited to expert opinion and at present, there is little evidence-based guidance on the timing of TPO-RA administration and on tapering/discontinuation of the drug. To provide clinical practice recommendations for these aspects, we used the Delphi technique to obtain consensus for statements regarding administration and on tapering/ discontinuation of second-line TPO-RAs among a group of Italian clinicians with expertise in management of ITP.

# **Materials and methods**

The Delphi process is a widely adopted technique that is used to achieve expert consensus. The method adopts a survey type format, where statements undergo successive rounds of voting until consensus is reached.<sup>21</sup> The Delphi approach thus combines the evidence-based medicine with an iterative and anonymous voting process. Moreover, the Delphi approach avoids problems in group dynamics as the experts can provide their opinions freely and anonymously.<sup>22</sup>

The steering committee was composed of four experts (FB, MC, SMS, FZ) who drafted five statements on administration and on tapering/ discontinuation of second-line TPO-RAs. Committee members were identified by criteria that included their expertise in management of ITP, publications, and attendance at international meetings.

An expert panel consisting of 11 Italian hematologists was invited to participate in the Delphi process based on their interest and expertise in the management of patients with ITP. The experts were identified from all geographic regions in order to be representative of clinical practice of ITP in Italy. The process used herein was composed of four steps: (a) establishment of a scientific steering committee who reviewed the relevant literature and developed the statements to be ranked; (b) first round of online voting by the expert panel; (c) modification and/or second round of voting for unmodified statements not reaching consensus by the steering committee based on comments by expert panel; (d) second round of online voting by the expert panel.

The experts were required to express their level of agreement with each statement, using a 5-point Likert-type scale (1 = disagree; 2 = somewhat disagree; 3 = neither agree nor disagree; 4 = somewhat agree; 5 = agree). Consensus was considered if either the sum of answers 1 and 2 (negative agreement), or 4 and 5 (positive agreement) exceeded 75% in line with previous studies with the Delphi method.<sup>23–26</sup> The results are expressed as a percentage of agreement with each statement.

# **Results and discussion**

The Delphi process described was used to vote upon five statements regarding initiation and on tapering/discontinuation of second-line TPO-RAs in patients with ITP (Table 1). Eleven experts participated in the voting. In the first round of voting, consensus was reached for

	Statement	% agreement
1	An early switch from corticosteroids to a TPO-RA has the dual advantage of sparing patients from corticosteroid abuse and improve long-term clinical outcomes.	100%
2A	Dose reduction (tapering) of TPO-RAs can be considered in patients with a stable response and platelet count $>50 \times 10^{\circ}$ /L (PR) that is maintained for at least 6 months in the absence of concomitant treatments.	54.5%
2B	Dose reduction (tapering) of TPO-RAs can be considered in patients with a stable response and platelet count $>50 \times 10^{\circ}$ /L (PR) that is maintained for at least 6 months in the absence of concomitant treatments.	63.6%
3	Dose reduction (tapering) of TPO-RAs can be considered in patients with a stable response and platelet count $>100 \times 10^{9}$ /L (CR) that is maintained for at least 6 months in the absence of concomitant treatments.	100%
4A	If TPO-RA treatment is given early, there is a greater chance of achieving partial or complete response.	72.7%
4B	Early treatment with a TPO-RA is associated with an increase in clinically significant response (partial or complete).	72.7%
5	Optimization of tapering and discontinuation of TPO-RA therapy in selected patients can improve the quality of life.	90.1%

**Table 1.** Statements on administration and on tapering/discontinuation of second-line TPO-RAs for treatment of ITP.

statements 1, 3, and 5. Statement 2 was voted upon again in the original form, while statement 4 was modified and voted upon again. Consensus was not reached for either.

Statement 1: An early switch from corticosteroids to a TPO-RA has the dual advantage of sparing patients from corticosteroid abuse and improve long-term clinical outcomes.

Unanimous consensus was reached on this statement. ASH guidelines recommend against a prolonged course of corticosteroids, stating that while there is possibly 'trivial benefit' an alternative therapy should be preferred.<sup>5</sup> Nevertheless, the 6weeks of corticosteroid treatment may be used to understand if the patient will enter remission or will need additional treatment. The guidelines further consider that prolonged (more than 6 weeks) administration of corticosteroids will be likely associated with an increased occurrence of adverse events. Indeed, the long-term side effects are well known and may involve many organs.<sup>27-30</sup> Moreover, since patients with ITP already have a high disease burden, long-term use of corticosteroids does not seem reasonable due to their well

known adverse events. However, the extended use of corticosteroids is common practice as highlighted in a real-world analysis on almost 8000 patients with ITP, wherein long-term, high-dose corticosteroid use was seen for up to 7 lines of therapy and frequently as monotherapy.<sup>31</sup> The expert panel fully agreed that the early use of a TPO-RA is a safe approach, can spare patients glucocorticoid toxicity, and, as such, improve long-term clinical outcomes.

Statement 2: Dose reduction (tapering) of TPO-RAs can be considered in patients with a stable response and platelet count  $>50 \times 10^{9}/L$  [partial response (PR)] that is maintained for at least 6 months in the absence of concomitant treatments.

Dose reduction (tapering) of TPO-RAs can be considered in patients with a stable response and platelet count  $>50 \times 10^9$ /L (PR) that is maintained for at least 6 months in the absence of concomitant treatments. We defined complete response (CR) as a platelet count  $>100 \times 10^9$ /L and response as a platelet count  $>30 \times 10^9$ /L and at least twofold increase in the baseline count as described in Rodeghiero *et al.*<sup>1</sup> Nonetheless, in recent years, many authors have considered a platelet response defined as a platelet count  $>50 \times 10^{9}/L$  at a given assessment on treatment with TPO-RA or placebo<sup>32</sup> or a target platelet range variable from  $50 \times 10^{9}/L$  and  $200 \times 10^{9}/L$ or higher.<sup>33</sup> This variability in definitions makes comparison among results in different studies difficult both in term of response during ongoing treatment and even more so during and after tapering of TPO-RAs.

No consensus was reached for statement 2 after two rounds of voting. The main reasons leading to this lack of consent were that (a) only a relatively small proportion of patients with a platelet count  $>50 \times 10^{9}/L$  will maintain response after discontinuation; and (b) the evidence was adequate for patients with CR, but not with PR. Other authors claimed that tapering can be attempted in patients with a platelet count  $>50 \times 10^9$ /L that is sustained for at least 6 months, with the caveat that the tapering can be interrupted if the platelet count drops below  $30 \times 10^{9}/L$ or  $50 \times 10^{9}$ /L if a bleeding event is documented.<sup>20</sup> These suggestions were based on the analysis of the published literature documenting delayed responses after discontinuation of a TPO-RA.<sup>20</sup> The trial by Cervinek reported on 46 patients being treated with a TPO-RA.34 In patients achieving a platelet count  $\geq 50 \times 10^9/L$  for 24 consecutive weeks, the drug was tapered over a period of 3 weeks. Eleven of the 46 patients discontinued the TPO-RA, and all maintained response for a median of 33 months. In the very recent trial by Lucchini et al.,35 13 of 51 (25%) patients considered as responders (platelet count  $\geq$  30  $\times$  10<sup>9</sup>/L and at least a twofold increase from the baseline count) were able to taper and discontinue eltrombopag and maintain remission for 24 weeks, while another 13 of the remaining 34 patients (38%) had started tapering. In addition, in both studies a proportion of patients who achieved a treatment-free response were also partial responders. Most recently, Cooper et al.<sup>36</sup> published a UK hematologist survey about tapering and discontinuation of TPO-RA in patients with ITP, stating that approximately 30% of patients are eligible for tapering and discontinuation, which may be considered after 6-12 months for patients demonstrating an adequate treatment response (platelet count  $>50,000/\mu$ L at  $\ge 75\%$  of assessments in the preceding 6 months).

Sustained remission off-treatment is possible in a proportion of patients. However, the available data in patients achieving a PR and discontinuing a TPO-RA are somewhat limited. Among the group of expert panelists, there was no agreement that discontinuation of the TPO-RA should be considered in patients with a PR.

Statement 3: Dose reduction (tapering) of TPO-RAs can be considered in patients with a stable response and platelet count  $>100 \times 10^9/L$  (CR) that is maintained for at least 6 months in the absence of concomitant treatments.

For reference, the main published studies on discontinuation of TPO-RAs in patients with ITP are summarized in Table 2. The experts agreed that dose reduction (tapering) of TPO-RAs can be considered in patients with a stable response and platelet count  $>100 \times 10^{9}/L$  (CR) that is maintained for at least 6 months in the absence of concomitant treatments. Gonzalez-Lopez reported on a cohort of 49 evaluable patients discontinuing eltrombopag; of these, 26 maintained a sustained response after a median of 9 months.<sup>39</sup> In the study by Marshall et al.,<sup>41</sup> 12 of 43 patients (28%) who discontinued romiplostim maintained an elevated platelet response after a median of more than 6 years, thus demonstrating that long-term offtreatment response can be attained. In the study by Mahevas et al.<sup>40</sup> on 54 patients, a total of 20 patients with CR discontinued TPO-RA and of these 8 maintained sustained response after a median of 13.5 months. Similar results have been reported by other authors, suggesting that discontinuation of a TPO-RA is associated with sustained platelet responses in a subset of patients who achieve a CR.42-44 Accordingly, tapering and discontinuation can be considered in these patients.

Statement 4: If TPO-RA treatment is given early, there is a greater chance of achieving partial or complete response.

Near consensus was reached after two rounds of voting for statement 4 regarding the possibility that early treatment is associated with higher rates of PR or CR. The agreement of 72.7% *versus* 75% needed for full consensus was obtained twice in two consecutive rounds of voting repeated on the same rephrased statement. In published studies, as noted in a recent review by Zaja *et al.*, CR or PR with TPO-RAs are more often observed

· .		•					
Study	Number of patients	Patients who discontinued TPO-RA ( <i>n</i> , % from all pts)	Patients in sustained remission (n, % of all patients)	Median follow-up (months)			
Newland <i>et al.</i> <sup>37</sup>	75	31 (41)	24 (32)	6			
Ghadaki <i>et al.</i> <sup>38</sup>	31	9 (29)	5 (16)	9.5			
González-Lopez et al. <sup>39</sup>	201	80 (39)	42 (21)	9			
Mahevas <i>et al.</i> <sup>40</sup>	54	20 (37)	8 (15)	13.5			
Cervinek et al. <sup>34</sup>	46	11 (24)	11 (24)	33			
Marshall <i>et al.</i> 41	43	12 (28)	12 (28)	33			
ITP, immune thrombocytopenia; TPO-RA, thrombopoietin receptor agonists.							

Table 2.	Main publishe	d studies on	discontinuation	of TPO-RAs in	patients with ITP.
----------	---------------	--------------	-----------------	---------------	--------------------

when initiated early, especially before splenectomy, or before early switch from first-line treatments.<sup>20,39,44,45</sup> These results might suggest that early use of a TPO-RA can be associated with improved responses, which would lead to better long-term outcomes.<sup>20</sup> In order to find predictors of remission, Newland et al.37 presented a pooled analysis of 13 studies involving 911 adults with ITP being treated with romiplostim; treatmentfree responses were observed in 61 patients. In multivariate analysis, shorter duration of ITP was significantly associated with a higher odds ratio of achieving a treatment-free response for both newly diagnosed and persistent ITP (<3 months *versus* >12 months, OR=4.275, p < 0.0001; 3–12 months versus > 12 months, OR = 2.171, p = 0.0408). Previous splenectomy and bleeding were not predictors of treatment-free response. Early use of TPO-RAs has also been observed in other studies. As reported by Gonzalez-Lopez in a study of 220 adults with ITP, PR, and CR rates were higher in those treated earlier.<sup>46</sup> The study by Lucchini et al.35 on eltrombopag mentioned in statement 2 enrolled patients with early-stage ITP (i.e. not responsive or in relapse after a full course of corticosteroid therapy and no secondline therapy). As suggested by the authors, TPO-RAs could help to maintain platelet counts until the immune system reestablishes an equilibrium, further supporting their early administration.35

Statement 5: Optimization of tapering and discontinuation of TPO-RA therapy in selected patients can improve the quality of life.

Consensus was reached for statement 5 that optimization of tapering and discontinuation of TPO-RA therapy in selected patients can improve the quality of life. While some studies have shown TPO-RAs can improve the patient's quality of life through reduction of bleeding events and emergency treatments,<sup>47,48</sup> there is limited data on the quality of life in patients who successfully discontinued a TPO-RA. However, it should be noted that among the goals of treatment, a positive effect on the quality of life should be a primary objective.<sup>3</sup>

A general improvement of OoL can be reasonably expected in patients who achieved a complete discontinuation of the drug. Nonetheless, even a simple reduction of the dosage (i.e. tapering of the dose) in many patients might be related to better tolerability of treatment and eventually to improvement of QoL. One panelist referred that the statement can be considered valid only in patients who achieve optimal levels of platelet counts, and that patients with lower levels may continue to feel insecure. Another commented that balance is needed between frequent clinical evaluations and effective long-term control of platelet counts. In addition, the common feeling was that the physician-patient relationship is important, as well as the patient's view, individual clinical history, and duration of disease. Finally, if the patient is rid of the burden of therapy with good disease control, his or her quality of life will be improved.

## Conclusions

The present Delphi method consolidates the current consensus on administration and tapering/ discontinuation of second-line TPO-RAs in patients with ITP. Specifically, the evidence regards the benefits of an early switch from corticosteroids to a TPO-RA and dose tapering of TPO-RAs in patients with a stable response and platelet count  $>100 \times 10^{9}$ /L for at least 6 months in the absence of concomitant treatments. For treatment discontinuation in patients with lower platelet counts ( $50 \times 10^{9}/L$ ), consensus was not achieved, and clinicians should decide according to the bleeding risk of each individual patient. It should be mentioned that the present recommendations are only applicable in countries where TPO-RAs are available at any stage in the treatment of ITP, although where available the recommendations could be useful to advocate early reimbursement of TPO-RA. Furthermore, the consensus highlighted areas where further studies are needed, namely the possibility to taper the TPO-RA in patients with PR and the potential benefits of early treatment with a TPO-RA in achieving clinically meaningful responses. Near consensus was obtained on the first issue, whereas on the second further studies are warranted.

## **Author contributions**

All authors contributed to the design of the research, to the analysis of the results and to the writing of the manuscript. MC and FZ discussed the results and commented on the manuscript.

## **Conflict of interest statement**

The authors declared the following potential conflicts of interest with respect to the research, authorship, and/or publication of this article: CM received honoraria from Amgen and Novartis for serving on advisory boards; SS received honoraria from CSL, AMGEN, Novartis, Novo Nordisk, SOBI, and Bayer; ZF received honoraria and funding from Novartis, Amgen, and Grifols; PF received honoraria from Novartis; FB received consultation honoraria from Amgen, Novartis, and Momenta; BC received honoraria from Novartis and Amgen; BA has received honoraria from Amgen, Novartis, Novo Nordisk, Bayer, and Takeda; GG, SPR, BE, CF, CG, RE, VA have no conflict of interest to declare.

## Funding

The authors disclosed receipt of the following financial support for the research, authorship, and/or publication of this article: Editorial support was provided by Content Ed Net, with the helpful assistance of Dr. Patrick Moore. The authors thank Novartis Farma Italy for funding the editorial support.

## ORCID iD

Monica Carpenedo D https://orcid.org/0000-0002-4355-3954

### References

- Rodeghiero F, Stasi R, Gernsheimer T, *et al.* Standardization of terminology, definitions and outcome criteria in immune thrombocytopenic purpura of adults and children: report from an international working group. *Blood* 2009; 113: 2386–2393.
- McMillan R, Bussel JB, George JN, *et al.* Selfreported health-related quality of life in adults with chronic immune thrombocytopenic purpura. *Am J Hematol* 2008; 83: 150–154.
- Provan D, Arnold DM, Bussel JB, et al. Updated international consensus report on the investigation and management of primary immune thrombocytopenia. Blood Adv 2019; 3: 3780–3817.
- Trotter P and Hill QA. Immune thrombocytopenia: improving quality of life and patient outcomes. *Patient Relat Outcome Meas* 2018; 9: 369–384.
- Neunert C, Terrell DR, Arnold DM, *et al.* American Society of Hematology 2019 guidelines for immune thrombocytopenia. *Blood Adv* 2019; 3: 3829–3866.
- Arnold DM. Positioning new treatments in the management of immune thrombocytopenia. *Pediatr Blood Cancer* 2013; 60(Suppl. 1): S19–S22.
- Neunert CE and Cooper N. Evidence-based management of immune thrombocytopenia: ASH guideline update. *Hematology Am Soc Hematol Educ Program* 2018; 2018: 568–575.
- Matzdorff A, Meyer O, Ostermann H, et al. Immune thrombocytopenia – current diagnostics and therapy: recommendations of a Joint Working Group of DGHO, OGHO, SGH, GPOH, and DGTI. Oncol Res Treat 2018; 41(Suppl. 5): 1–30.
- Mingot-Castellano ME, Roman MTA, Fernandez Fuertes LF, et al. Management of adult patients with primary immune thrombocytopenia (ITP) in clinical practice: a consensus approach of the Spanish ITP Expert Group. Adv Hematol 2019; 2019: 4621416.
- 10. Cooper N, Kruse A, Kruse C, *et al.* Immune thrombocytopenia (ITP) World Impact Survey

(iWISh): patient and physician perceptions of diagnosis, signs and symptoms, and treatment. *Am J Hematol* 2021; 96: 188–198.

- 11. Kubasch AS, Kisro J, Hessling J, *et al.* Disease management of patients with immune thrombocytopenia-results of a representative retrospective survey in Germany. *Ann Hematol* 2020; 99: 2085–2093.
- Reid MM. Chronic idiopathic thrombocytopenic purpura: incidence, treatment, and outcome. *Arch Dis Child* 1995; 72: 125–128.
- Stasi R, Stipa E, Masi M, et al. Long-term observation of 208 adults with chronic idiopathic thrombocytopenic purpura. Am J Med 1995; 98: 436–442.
- Ghanima W, Godeau B, Cines DB, *et al.* How I treat immune thrombocytopenia: the choice between splenectomy or a medical therapy as a second-line treatment. *Blood* 2012; 120: 960–969.
- 15. Cuker A. Transitioning patients with immune thrombocytopenia to second-line therapy: challenges and best practices. *Am J Hematol* 2018; 93: 816–823.
- 16. Bowdler AJ. *The complete spleen: structure, function and clinical significance.* New York: Humana Press, 2001.
- Chaturvedi S, Arnold DM and McCrae KR. Splenectomy for immune thrombocytopenia: down but not out. *Blood* 2018; 131: 1172–1182.
- Lucchini E, Zaja F and Bussel J. Rituximab in the treatment of immune thrombocytopenia: what is the role of this agent in 2019. *Haematologica* 2019; 104: 1124–1135.
- Pavord S, Thachil J, Hunt BJ, et al. Practical guidance for the management of adults with immune thrombocytopenia during the COVID-19 pandemic. Br J Haematol 2020; 189: 1038–1043.
- Zaja F, Carpenedo M, Barate C, *et al.* Tapering and discontinuation of thrombopoietin receptor agonists in immune thrombocytopenia: real-world recommendations. *Blood Rev* 2020; 41: 100647.
- Loblaw DA, Prestrud AA, Somerfield MR, et al. American Society of Clinical Oncology clinical practice guidelines: formal systematic reviewbased consensus methodology. J Clin Oncol 2012; 30: 3136–3140.
- Bellizzi V, Bianchi S, Bolasco P, et al. A Delphi consensus panel on nutritional therapy in chronic kidney disease. J Nephrol 2016; 29: 593–602.
- 23. Blasi F, Concia E, Del Prato B, *et al.* The most appropriate therapeutic strategy for acute lower

respiratory tract infections: a Delphi-based approach. *J Chemother* 2017; 29: 274–286.

- 24. Brown AK, O'Connor PJ, Roberts TE, et al. Recommendations for musculoskeletal ultrasonography by rheumatologists: setting global standards for best practice by expert consensus. *Arthritis Rheum* 2005; 53: 83–92.
- 25. Mahler DA, Selecky PA, Harrod CG, *et al.* American College of Chest Physicians consensus statement on the management of dyspnea in patients with advanced lung or heart disease. *Chest* 2010; 137: 674–691.
- Zafar SY, Currow DC, Cherny N, *et al.* Consensus-based standards for best supportive care in clinical trials in advanced cancer. *Lancet Oncol* 2012; 13: e77–e82.
- Caplan A, Fett N, Rosenbach M, et al. Prevention and management of glucocorticoidinduced side effects: a comprehensive review: a review of glucocorticoid pharmacology and bone health. J Am Acad Dermatol 2017; 76: 1–9.
- Oray M, Abu Samra K, Ebrahimiadib N, et al. Long-term side effects of glucocorticoids. Expert Opin Drug Saf 2016; 15: 457–465.
- 29. Rhen T and Cidlowski JA. Antiinflammatory action of glucocorticoids – new mechanisms for old drugs. *N Engl J Med* 2005; 353: 1711–1723.
- Schacke H, Docke WD and Asadullah K. Mechanisms involved in the side effects of glucocorticoids. *Pharmacol Ther* 2002; 96: 23–43.
- Bussel JB, Tkacz J, Manjelievskaia J, et al. Longterm overuse of corticosteroids in patients with immune thrombocytopenia: a real-world analysis of a US Claims Database. In: Poster presented at ASH 2020, 5–8 December 2020, https://ash. confex.com/ash/2020/webprogram/Paper133757. html
- Al-Samkari H and Kuter DJ. Optimal use of thrombopoietin receptor agonists in immune thrombocytopenia. *Ther Adv Hematol* 2019; 10: 2040620719841735.
- Kuter DJ, Tarantino MD and Lawrence T. Clinical overview and practical considerations for optimizing romiplostim therapy in patients with immune thrombocytopenia. *Blood Rev* 2021; 49: 100811.
- 34. Cervinek L, Mayer J and Doubek M. Sustained remission of chronic immune thrombocytopenia after discontinuation of treatment with thrombopoietin-receptor agonists in adults. *Int J Hematol* 2015; 102: 7–11.

- 35. Lucchini E, Palandri F, Volpetti S, et al. Eltrombopag second-line therapy in adult patients with primary immune thrombocytopenia in an attempt to achieve sustained remission off-treatment: results of a phase II, multicentre, prospective study. Br J Haematol 2021; 193: 386–396.
- Cooper N, Hill QA, Grainger J, et al. Tapering and discontinuation of thrombopoietin receptor agonist therapy in patients with immune thrombocytopenia: results from a modified Delphi panel. Acta Haematol 2021; 144: 418–426.
- Newland A, Bussel JB, Bird R, et al. Predictors of remission in adults with immune thrombocytopenia treated with romiplostim. *Blood* 2018; 132: 735–735.
- 38. Ghadaki B, Nazi I, Kelton JG, *et al.* Sustained remissions of immune thrombocytopenia associated with the use of thrombopoietin receptor agonists. *Transfusion* 2013; 53: 2807–2812.
- Gonzalez-Lopez TJ, Pascual C, Alvarez-Roman MT, et al. Successful discontinuation of eltrombopag after complete remission in patients with primary immune thrombocytopenia. Am J Hematol 2015; 90: E40–E43.
- Mahevas M, Fain O, Ebbo M, et al. The temporary use of thrombopoietin-receptor agonists may induce a prolonged remission in adult chronic immune thrombocytopenia. Results of a French observational study. Br J Haematol 2014; 165: 865–869.
- 41. Marshall AL, Scarpone R, De Greef M, *et al.* Remissions after long-term use of romiplostim for immune thrombocytopenia. *Haematologica* 2016; 101: e476–e478.

- 42. Bussel JB, Wang X, Lopez A, *et al.* Case study of remission in adults with immune thrombocytopenia following cessation of treatment with the thrombopoietin mimetic romiplostim. *Hematology* 2016; 21: 257–262.
- Carpenedo M, Cantoni S, Coccini V, *et al.* Feasibility of romiplostim discontinuation in adult thrombopoietin-receptor agonist responsive patients with primary immune thrombocytopenia: an observational retrospective report in real life clinical practice. *Hematol Rep* 2015; 7: 5673.
- 44. Mazzucconi MG, Santoro C, Baldacci E, *et al.* TPO-RAs in pITP: description of a case series and analysis of predictive factors for response. *Eur J Haematol* 2017; 98: 242–249.
- 45. Wong RSM, Saleh MN, Khelif A, et al. Safety and efficacy of long-term treatment of chronic/ persistent ITP with eltrombopag: final results of the EXTEND study. *Blood* 2017; 130: 2527– 2536.
- Gonzalez-Lopez TJ, Fernandez-Fuertes F, Hernandez-Rivas JA, et al. Efficacy and safety of eltrombopag in persistent and newly diagnosed ITP in clinical practice. Int J Hematol 2017; 106: 508–516.
- 47. George JN, Mathias SD, Go RS, *et al.* Improved quality of life for romiplostim-treated patients with chronic immune thrombocytopenic purpura: results from two randomized, placebo-controlled trials. *Br J Haematol* 2009; 144: 409–415.
- Khelif A, Saleh MN, Salama A, et al. Changes in health-related quality of life with long-term eltrombopag treatment in adults with persistent/ chronic immune thrombocytopenia: findings from the EXTEND study. Am J Hematol 2019; 94: 200–208.

Visit SAGE journals online journals.sagepub.com/ home/tah

SAGE journals